Author's response to reviews

Title: Comparing sexual risks and patterns of alcohol and drug use between injection drug users (IDUs) and non-IDUs who report sexual partnerships with IDUs in St. Petersburg, Russia

Authors:

Nadia Abdala (nadia.abdala@yale.edu)
Edward White (e.white@yale.edu)
Olga V Tousova (tousova@biomed.spb.ru)
Tatiana V Krasnoselskikh (tatiana.krasnoselskikh@gmail.com)
Sergei Verevochkin (sverev@biomed.spb.ru)
Andrei P Kozlov (contact@biomed.spb.ru)
Robert Heimer (robert.heimer@yale.edu)

Version: 3 Date: 2 November 2010

Author's response to reviews: see over
November 01, 2010

Mr. Jack Cochrane  
The Biomed Central Editorial Team  
BioMed Central Ltd  
Floor 6  
236 Gray's Inn Road  
London WC1X 8HB  
United Kingdom

RE: “Comparing sexual risks and patterns of alcohol and drug use between injection drug users (IDUs) and non-IDUs who report sexual partnerships with IDUs in St Petersburg, Russia”. Manuscript number 9824491374082121.

Dear Mr. Cochrane:

Thank you for accepting our manuscript for publication. According to your request we made the following formatting changes to the final version of the manuscript:
- Context information was added to the background section of the abstract
- The funding information was deleted from the title page
- The vertical lines were removed from the tables
- The reference section was renamed to “References”
- A check for typographical errors was conducted

We thank the editors for their consideration of our revised manuscript.

Sincerely,

Nadia Abdala, PhD  
Yale School of Public Health  
60 College Street, 507 LEPH  
New Haven, CT 06520 - USA  
phone: 203-785-2747  
fax: 203-785-3260  
nadia.abdala@yale.edu